\documentclass[12pt,reqno]{article}

\usepackage[usenames]{color}
\usepackage{amssymb}
\usepackage{amsmath}
\usepackage{amsthm}
\usepackage{amsfonts}
\usepackage{amscd}
\usepackage{graphicx}

\usepackage[colorlinks=true,
linkcolor=webgreen,
filecolor=webbrown,
citecolor=webgreen]{hyperref}

\definecolor{webgreen}{rgb}{0,.5,0}
\definecolor{webbrown}{rgb}{.6,0,0}

\usepackage{color}
\usepackage{fullpage}
\usepackage{float}

\usepackage{graphics}
\usepackage{latexsym}
\usepackage{epsf}
\usepackage{breakurl}

\usepackage{multirow}
\usepackage{pgf,tikz,pgfplots}
\usepackage{float}
\pgfplotsset{compat=1.12}
\usetikzlibrary{arrows}
\usetikzlibrary{positioning}
\definecolor{colAp1}{rgb}{1,0,0}
\definecolor{colAp2}{rgb}{0,0.2,0.8}
\definecolor{colAp3}{rgb}{0,0.6,0}
\DeclareMathOperator{\lcm}{lcm}
\def\ts{\textstyle}
\def\lf{\lfloor}
\def\rf{\rfloor}
\def\Iff{\Leftrightarrow}
\def\LIff{\Longleftrightarrow}
\def\la{\langle}
\def\ra{\rangle}
\def\A{\mathcal A}
\def\N{\mathbb N}
\def\Z{\mathbb Z}
\def\d{\delta}
\def\l{\lambda}
\def\m{\textbf{m}}
\def\sm{\setminus}

\DeclareMathOperator{\Ap}{Ap}
\DeclareMathOperator{\APP}{AP}
\DeclareMathOperator{\FG}{FG}

\setlength{\textwidth}{6.5in}
\setlength{\oddsidemargin}{.1in}
\setlength{\evensidemargin}{.1in}
\setlength{\topmargin}{-.1in}
\setlength{\textheight}{8.4in}

\newcommand{\seqnum}[1]{\href{https://oeis.org/#1}{\rm \underline{#1}}}

\begin{document}
\begin{thebibliography}{99}

\bibitem{Bat58}
P. T. Bateman, Remark on a recent note on linear forms, {\it Amer. Math. Monthly} {\bf 65} (1958), 517--518.

\bibitem{BS62}
A. Brauer and J. E. Shockley, On a problem of Frobenius, {\it J. Reine Angew. Math.} {\bf 211} (1962), 215--220.

\bibitem{Gra73}
D. D. Grant, On linear forms whose coefficients are in arithmetic progression, {\it Israel J. Math.} {\bf 15} (1973), 204--209.

\bibitem{Joh60}
S. M. Johnson, A linear diophantine problem, {\it Canad. J. Math.} {\bf 12} (1960), 390--398.

\bibitem{KNP16}
C. Kiers, C. O'Neill, and V. Ponomarenko, Numerical semigroups on compound sequences, {\it Comm. Algebra} {\bf 44} (2016), 3842--3852. 

\bibitem{Rob56}
J. B. Roberts, Note on linear forms, {\it Proc. Amer. Math. Soc.} {\bf 7} (1956), 465--469.

\bibitem{Ros05}
J. C. Rosales, Fundamental gaps of numerical semigroups generated by two elements, {\it Linear Algebra Appl.} {\bf 405} (2005), 200--208. 

\bibitem{RG-S04}
J. C. Rosales and P. A. Garc\'{i}a-S\'{a}nchez, Numerical semigroups
with embedding dimension three, {\it Arch. Math.} {\bf 83} (2004), 488--496.

\bibitem{RGGM04}
J. C. Rosales, P. A. Garc\'{i}a-S\'{a}nchez, J. I. Garc\'{i}a-Garc\'{i}a, and J. A. Jim\'{e}nez Madrid, Fundamental gaps in numerical semigroups, {\it J. Pure Appl. Algebra} {\bf 189} (2004), 301--313. 

\bibitem{Sel77}
E. S. Selmer, On the linear diophantine problem of Frobenius, {\it J. Reine Angew. Math.} {\bf 293/294} (1977), 1--17.

\bibitem{Tri94}
A. Tripathi, The coin exchange problem for arithmetic progressions, {\it Amer. Math. Monthly} {\bf 101} (1994), 779--781.

\bibitem{Tri03}
A. Tripathi, On a variation of the coin exchange problem for arithmetic progressions, {\it Integers} {\bf 3} (2003), Article A01.

\bibitem{Tri06}
A. Tripathi, On a linear diophantine problem of Frobenius, {\it Integers} {\bf 6} (2006), Article A14.

\bibitem{Tri13}
A. Tripathi, A note on a special case of the Frobenius problem, {\it Indian J. Pure Appl. Math.} {\bf 44} (2013), 375--381.  

\bibitem{Tri17}
A. Tripathi, On a special case of the Frobenius problem, {\it J. Integer
Sequences} {\bf 20} (2017), 
\href{https://cs.uwaterloo.ca/journals/JIS/VOL20/Tripathi/trip40.html}{Article 17.7.2}.

\bibitem{TV06}
A. Tripathi and S. Vijay, On a generalization of the coin exchange
problem for three variables,  {\it J. Integer Sequences} {\bf 9} (2006),
\href{https://cs.uwaterloo.ca/journals/JIS/VOL9/Tripathi2/tripathi24.html}{Article 06.4.6}.

\end{thebibliography}


\begin{center}
\includegraphics[width=4in]{logo129.eps}
\end{center}

\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}


\begin{center}
\vskip 1cm{\LARGE\bf  Some Results on Fundamental Gaps in Numerical Semigroups\\
\vskip 1cm}
Edgar Federico Elizeche and
Amitabha Tripathi\footnote{Corresponding author.}\\
Department of Mathematics\\
Indian Institute of Technology\\
Hauz Khas\\
New Delhi 110016\\
India\\
\href{mailto:maz188235@maths.iitd.ac.in}{\tt maz188235@maths.iitd.ac.in} \\
\href{mailto:atripath@maths.iitd.ac.in}{\tt atripath@maths.iitd.ac.in}
\end{center}

\vskip .2 in

\begin{abstract}
A numerical semigroup is a submonoid of ${\Z}_{\ge 0}$ whose complement
in ${\Z}_{\ge 0}$ is finite. The gap set $\text{G}(S)$ of a numerical
semigroup $S$ is the finite set ${\Z}_{\ge 0} \sm S$. A positive integer
$n$ is in the set $\FG(S)$ of fundamental gaps of $S$ provided
$n \notin S$ but $kn \in S$ for each $k \in \Z$, $k>1$. We explore the
set $\FG(S)$ mostly when $S$ is generated by two or three integers,
but also in some other special cases, including when $S$ is generated
by arithmetic progressions.
\end{abstract}

\section{Introduction} \label{intro}

A \textit{numerical semigroup\/} $S$ is a submonoid of ${\Z}_{\ge 0}$ whose complement ${\Z}_{\ge 0} \sm S$ is finite. For the complement to be finite, it is necessary and sufficient that $\gcd(S)=1$. For a given subset $A$ of positive integers, we write
\[ \la A \ra = \big\{ a_1x_1+\cdots+a_kx_k: a_i \in A, x_i \in {\Z}_{\ge 0}, k \in \N \big\}. \]
Note that $\la A \ra$ is a submonoid of ${\Z}_{\ge 0}$, and that $S=\la A \ra$ is a numerical semigroup if and only if $\gcd(A)=1$.  

We say that $A$ is a set of generators of the semigroup $S$, or that the semigroup $S$ is generated by the set $A$, when $S=\la A \ra$. Further, $A$ is a minimal set of generators for $S$ if $A$ is a set of generators of $S$ and no proper subset of $A$ generates $S$. Every semigroup has a unique minimal set of generators. The embedding dimension $e(S)$ of $S$ is the size of the minimal set of generators. 

A very useful tool in the study of numerical semigroups is the determination of an Ap\'{e}ry set of the semigroup. Given a numerical semigroup $S$, and $a \in S$, the Ap\'{e}ry set of $S$ corresponding to $a$ is given by 
\[ \Ap(S,a) = \left\{ {\m}_x: x \in \{0,1,2,\ldots,a-1\} \right\}, \]  
where ${\m}_x$ denotes the least positive integer in $S$ congruent to $x$ modulo $a$.  

The finite set ${\Z}_{\ge 0} \sm S$ is called the 
\textit{gap set\/} of $S$, and is denoted by $\text{G}(S)$. If $n \in \text{G}(S)$ and $d \mid n$, $d \in \N$, then $d \in \text{G}(S)$. This naturally leads to the \textit{set $\FG(S)$ of fundamental gaps\/} of $S$, defined as 
\[  \FG(S) = \left\{ n \in \text{G}(S): kn \in S \;\forall\; k>1 \right\}; \]
see Rosales et al.\ \cite{RGGM04} for more details. 
 
We describe the set $\FG(S)$ in terms of the elements in $\Ap(S,a)$, for any $a \in S$, in Lemma~\ref{basic}.     

\begin{lemma} \label{basic}
Let $A$ be any set of positive integers with $\gcd(A)=1$, and let $S=\la
A \ra$ be the numerical semigroup generated by $A$. Let $a \in A$, and
let ${\m}_x$ denote the least positive integer in $S$ congruent to $x$
modulo $a$. Then $n \in \FG(S)$ if and only if
\[ n = {\m}_n - \l a, \quad 1 \le \l \le \ts\frac{1}{a} \min \left\{ {\m}_n - \ts\frac{1}{2} {\m}_{2n}, {\m}_n - \ts\frac{1}{3} {\m}_{3n} \right\}. \] 
\end{lemma}

\begin{proof}
Note that $n \in \FG(S)$ if and only if $n<{\m}_n$, $2n \ge {\m}_{2n}$, and $3n \ge {\m}_{3n}$. Thus, $n={\m}_n-\l a$ for some $\l \ge 1$. Using this in the other two constraints gives the upper bounds $\l \le \frac{1}{a}\big({\m}_n - \frac{1}{2}{\m}_{2n}\big)$ and $\l \le \frac{1}{a}\big({\m}_n - \frac{1}{3}{\m}_{3n}\big)$.  
\end{proof}

We determine the set of fundamental gaps of some numerical semigroups in
this article. We consider numerical semigroups with embedding dimension
$2$ in Section \ref{ed=2}. The Ap\'{e}ry set in this case is well known,
and we use this to determine the set of fundamental gaps, giving a simpler
proof of the same result by Rosales \cite{Ros05}.  We consider numerical
semigroups with embedding dimension $3$ in Section \ref{ed=3}. The
Ap\'{e}ry set in this case is in general difficult to compute. We consider
several special cases where the Ap\'{e}ry set has been determined, and
use that to determine the set of fundamental gaps in those cases. We
consider numerical semigroups generated by arithmetic progressions in
Section \ref{section=AP}. Ap\'{e}ry sets for such semigroups have been
determined, and we use these to determine the set of fundamental gaps.

\section{The case of embedding dimension $2$} \label{ed=2}

Numerical semigroups with embedding dimension $2$ are the simplest to study. The Ap\'{e}ry set of these semigroups is easy to see and part of basic number theory. Rosales \cite{Ros05} determined the fundamental gap of numerical semigroups $S=\la a,b \ra$, $\gcd(a,b)=1$, by making use of the well known fact that $n \notin S$ if and only if $n=ab-ax-by$ with $x,y \in \N$. We use the Ap\'{e}ry set of $S$ given in Lemma \ref{minclass_ab} to determine the set of fundamental gaps in $S$ in Theorem \ref{FG_two_var}. 

\begin{lemma} \label{minclass_ab}
Let $A=\{a,b\}$, where $\gcd(a,b)=1$. Then the Ap\'{e}ry set for the semigroup $S=\langle a,b \rangle$ is given by 
\[ \Ap(S,a) = \big\{ bx: 0 \le x \le a-1 \big\}. \]
\end{lemma}

\begin{theorem} {\bf \cite[Theorem 9]{Ros05}} \label{FG_two_var}
Let $S=\la a,b \ra$, where $\gcd(a,b)=1$. The set of fundamental gaps is given by 
\[ \FG(S) = \big\{ bs-ar: 1 \le r \le \ts\frac{b}{3}, \ts\frac{a}{2} \le s < \frac{2a}{3} \big\} \:\bigcup\: \big\{ bs-ar: 1 \le r \le \ts\frac{b}{2}, \ts\frac{2a}{3} \le s \le a-1 \big\}. \]
\end{theorem}

\begin{proof}
Recall that $n=bs-ar \notin \la a,b \ra$ if and only if $1 \le s \le a-1$ and $1 \le r<\frac{bs}{a}$, and that $n$ is a fundamental gap if and only if $n \notin \la a,b \ra$ and $2n, 3n \in \la a,b \ra$. 

Since $2n=2bs-2ar \notin \la a,b \ra$ if $s<\frac{a}{2}$, we may henceforth assume $s \ge \frac{a}{2}$. If $a$ is even and $s=\frac{a}{2}$, then $2n=a(b-2r) \in \la a,b \ra$ and $3n=b \cdot \frac{a}{2}+a(b-3r) \in \la a,b \ra$ if and only if $r \le \frac{b}{3}$. Hence $n=b \cdot \frac{a}{2}-ar$ is a fundamental gap if and only if $1 \le r \le \frac{b}{3}$.  

For the rest of this proof, suppose $s>\frac{a}{2}$. Now $2n=b(2s-a)+a(b-2r) \in \la a,b \ra$ if and only if $r \le \frac{b}{2}$, since $2s-a<a$. To decide whether or not $3n \in \la a,b \ra$, we consider the two cases (i) $s<\frac{2a}{3}$, and (ii) $\frac{2a}{3} \le s \le a-1$. In case (i),  $3n=b(3s-a)+a(b-3r) \in \la a,b \ra$ if and only if $r \le \frac{b}{3}$, since $0<3s-a<a$. In case (ii), $3n=b(3s-2a)+a(2b-3r) \in \la a,b \ra$ if and only if $r \le \frac{2b}{3}$, since $0<3s-2a<a$. Thus, in case (i), we have $r \le \min\big\{\frac{b}{2},\frac{b}{3}\big\}=\frac{b}{3}$, while in case (ii), we have $r \le \min\big\{\frac{b}{2},\frac{2b}{3}\big\}=\frac{b}{2}$.  
\end{proof}

\section{The case of embedding dimension $3$} \label{ed=3}

Numerical semigroups with embedding dimension $3$ have received a lot of attention, primarily because these are the first class of numerical semigroups that pose a challenge. By contrast to the case of numerical semigroups with embedding dimension $2$, Ap\'{e}ry sets of numerical semigroups with embedding dimension $3$ are usually quite difficult to describe. The Frobenius number $\text{F}(S)=\max (\N \sm S)$ of a numerical semigroup $S$ is easily computed from the Ap\'{e}ry set of $S$, since $\text{F}(S)=\max (\Ap(S,a))-a$ for each $a \in S$. Johnson \cite{Joh60} described an algorithm to compute $\text{F}(S)$ in terms of minimum multiples of generators that are in the numerical semigroup generated by the other two elements, but without determining the Ap\'{e}ry set. Rosales and Garc\'{i}a-S\'{a}nchez \cite{RG-S04} and independently, Tripathi and Vijay \cite{TV06}, were able to describe the Ap\'{e}ry set in terms of the constants first determined by Johnson \cite{Joh60}; see Proposition \ref{matrix}. However, the constants determined by Johnson \cite{Joh60} do not lead to an algebraic expression in terms of the generators of $S$ and, in particular, to the determination of the Ap\'{e}ry set of $S$ in the desired manner.  

In this section, we consider three cases of numerical semigroups with embedding dimension $3$ in which the Ap\'{e}ry sets can be computed explicitly in terms of their generators.\hfil


\begin{figure}[H]
\centering
\begin{tikzpicture}[line cap=round,line join=round,>=triangle 45,x=0.5cm,y=0.5cm]
\begin{axis}[
x=0.5cm,y=0.5cm,
axis lines=middle,
xmin=0,
xmax=25,
ymin=0,
ymax=23,
xticklabels={},
yticklabels={},
]
\end{axis}
   
\clip(-3,-3) rectangle (32.86830172000001,22.97239588);
\fill[line width=2pt,color=colAp1,fill=colAp1,fill opacity=0.10000000149011612] (0,7) -- (0,0) -- (3,0) -- (3,7) -- cycle;
\fill[line width=2pt,color=colAp1,fill=colAp1,fill opacity=0.10000000149011612] (0,4) -- (0,0) -- (8,0) -- (8,4) -- cycle;
\fill[line width=2pt,color=colAp2,fill=colAp2,fill opacity=0.1] (0,14) -- (0,0) -- (6,0) -- (6,14) -- cycle;
\fill[line width=2pt,color=colAp2,fill=colAp2,fill opacity=0.1] (0,8) -- (0,0) -- (16,0) -- (16,8) -- cycle;
\fill[line width=2pt,color=colAp3,fill=colAp3,fill opacity=0.1] (0,21) -- (0,0) -- (9,0) -- (9,21) -- cycle;
\fill[line width=2pt,color=colAp3,fill=colAp3,fill opacity=0.1] (0,12) -- (24,12) -- (24,0) -- (0,0) -- cycle;

\draw [line width=2pt,color=black,dashed] (0,23)-- (23,0);
\draw [line width=2pt,color=black] (0,25)-- (25,0);

\foreach \x in {0,1,...,10}{
        \draw [fill=red] (2*\x,23-2*\x) circle (2pt);
        \draw [fill=red] (2*\x,25-2*\x) circle (2pt);
                                          }
\draw [line width=2pt] (0,7)-- (0,0);
\draw [line width=2pt] (0,0)-- (3,0);
\draw [line width=2pt,color=colAp1] (0,7)-- (0,0);
\draw [line width=2pt,color=colAp1] (0,0)-- (3,0);
\draw [line width=2pt,color=colAp1] (3,0)-- (3,7);
\draw [line width=2pt,color=colAp1] (3,7)-- (0,7);
\draw [line width=2pt,color=colAp1] (0,4)-- (0,0);
\draw [line width=2pt,color=colAp1] (0,0)-- (8,0);
\draw [line width=2pt,color=colAp1] (8,0)-- (8,4);
\draw [line width=2pt,color=colAp1] (8,4)-- (0,4);
\draw [line width=2pt,color=colAp2] (0,14)-- (0,0);
\draw [line width=2pt,color=colAp2] (0,0)-- (6,0);
\draw [line width=2pt,color=colAp2] (6,0)-- (6,14);
\draw [line width=2pt,color=colAp2] (6,14)-- (0,14);
\draw [line width=2pt,color=colAp2] (0,8)-- (0,0);
\draw [line width=2pt,color=colAp2] (0,0)-- (16,0);
\draw [line width=2pt,color=colAp2] (16,0)-- (16,8);
\draw [line width=2pt,color=colAp2] (16,8)-- (0,8);
\draw [line width=2pt,color=colAp3] (0,21)-- (0,0);
\draw [line width=2pt,color=colAp3] (0,0)-- (9,0);
\draw [line width=2pt,color=colAp3] (9,0)-- (9,21);
\draw [line width=2pt,color=colAp3] (9,21)-- (0,21);
\draw [line width=2pt,color=colAp3] (0,12)-- (24,12);
\draw [line width=2pt,color=colAp3] (24,12)-- (24,0);
\draw [line width=2pt,color=colAp3] (24,0)-- (0,0);
\draw [line width=2pt,color=colAp3] (0,0)-- (0,12);
\draw [fill=black] (0,0) circle (2pt);
\draw[color=black] (-0.7,-0.7038741100000017) node {\rotatebox{45}{$(0,0)$}};
\draw [fill=colAp1] (0,4) circle (2.5pt);
\draw[color=colAp1] (-1.5,4) node {$(0,x_{13})$};
\draw [fill=colAp1] (0,7) circle (2.5pt);
\draw[color=colAp1] (-1.5,7) node {$(0,L_3)$};
\draw [fill=colAp1] (3,7) circle (2pt);
\draw[color=colAp1] (3.283233020000005,7.5) node {$(x_{12},L_3)$};
\draw [fill=colAp1] (8,4) circle (2pt);
\draw[color=colAp1] (7.7,4.5) node {$(L_2,x_{13})$};
\draw [fill=colAp1] (8,0) circle (2.5pt);
\draw[color=colAp1] (8,-1.3) node {\rotatebox{90}{$(L_2,0)$}};
\draw [fill=colAp1] (3,0) circle (2.5pt);
\draw[color=colAp1] (3,-1.4) node {\rotatebox{90}{$(x_{12},0)$}};
\draw [fill=colAp2] (0,8) circle (2.5pt);
\draw[color=colAp2] (-1.5,8) node {$(0,2x_{12})$};
\draw [fill=colAp2] (0,14) circle (2.5pt);
\draw[color=colAp2] (-1.5,14) node {$(0,2L_3)$};
\draw [fill=colAp2] (6,14) circle (2pt);
\draw[color=colAp2] (6.294886720000005,14.699206010000001) node {$(2x_{12},2L_3)$};
\draw [fill=colAp2] (6,0) circle (2.5pt);
\draw[color=colAp2] (6,-1.5) node {\rotatebox{90}{$(2x_{12},0)$}};
\draw [fill=colAp2] (16,0) circle (2.5pt);
\draw[color=colAp2] (16,-1.5) node {\rotatebox{90}{$(2L_2,0)$}};
\draw [fill=colAp2] (16,8) circle (2pt);
\draw[color=colAp2] (16.28649076000001,8.67589861) node {$(2L_2,2x_{13})$};
\draw [fill=colAp3] (0,12) circle (2.5pt);
\draw[color=colAp3] (-1.5,12) node {$(0,3x_{12})$};
\draw [fill=colAp3] (0,21) circle (2.5pt);
\draw[color=colAp3] (-1.5,21) node {$(0,3L_3)$};
\draw [fill=colAp3] (9,0) circle (2pt);
\draw[color=colAp3] (9,-1.5) node {\rotatebox{90}{$(3x_{12},0)$}};
\draw [fill=colAp3] (24,0) circle (2pt);
\draw[color=colAp3] (24,-1.5) node {\rotatebox{90}{$(3L_2,0)$}};
\draw [fill=colAp3] (9,21) circle (2pt);
\draw[color=colAp3] (9.271109200000007,21.67915635) node {$(3x_{12},3L_3)$};
\draw [fill=colAp3] (24,12) circle (2pt);
\draw[color=colAp3] (24.29394648000001,12.679626470000002) node {$(3L_2,3x_{13})$};
\end{tikzpicture}
\caption{
A geometric depiction of the set of fundamental gaps in numerical
semigroups with embedding dimension $3$.  Each lattice point $(x,y)$
represents the integer $n=a_2x+a_3y$. The integers in the Ap\'ery
set $\A$ (respectively, in $2\A$ and $3\A$) are those represented by
lattice points inside the region given by the union of two rectangles
enclosed by {\bf \textcolor{colAp1}{red}} lines (respectively, by {\bf
\textcolor{colAp2}{blue}} lines and by {\bf \textcolor{colAp3}{green}}
lines). The solid {\bf \textcolor{black}{black}} line represents the
equation $a_2x+a_3y=n$, whereas the dashed {\bf \textcolor{black}{black}}
line represents the equation $a_2x+a_3y=n-a_1$.  }
\end{figure}

\bigskip

\noindent We then use this to determine the set of fundamental gaps in each case. We deal with $3$-term compound sequences in Subsection \ref{CS}. These are given by $c_0=a_1a_2$, $c_1=b_1a_2$, $c_2=b_1b_2$, where $a_1,a_2$ and $b_1,b_2$ are pairs of positive integers such that $a_i<b_i$ for each $i$ and $\gcd(a_i,b_j)=1$ for each pair $i,j$, $i \ge j$. We consider numerical semigroups generated by $a,b,c$, where $a \mid (b+c)$, in Subsection \ref{b+c}. We consider numerical semigroups generated by $a,b,c$, where $a \mid \lcm(b,c)$, in Subsection \ref{lcm}.

\newpage

\begin{proposition} {\bf \cite{Joh60,RG-S04,TV06}} \label{matrix} 
Let $A=\{a_1,a_2,a_3\}$ be a set of positive integers, with $\gcd(a_1,a_2,a_3)=1$. Define $L_1, L_2, L_3$ by 
\begin{align*}
L_1 & = \min\big\{ k_1 \in \N: k_1a_1 = v_{12}a_2+v_{13}a_3,\: v_{12} \ge 0, v_{13} \ge 0 \big\}, \\
L_2 & = \min\big\{ k_2 \in \N: k_2a_2 = v_{21}a_1+v_{23}a_3,\: v_{21} \ge 0, v_{23} \ge 0 \big\}, \\ 
L_3 & = \min\big\{ k_3 \in \N: k_3a_3 = v_{31}a_1+v_{32}a_2,\: v_{31} \ge 0, v_{32} \ge 0 \big\}.
\end{align*}
Then there exist nonnegative integers $x_{12}, x_{13}, x_{21}, x_{23}, x_{31}, x_{32}$ such that
\[ L_1a_1 = x_{12}a_2 + x_{13}a_3, \quad L_2a_2 = x_{21}a_1 + x_{23}a_3, \quad L_3a_3 = x_{31}a_1 + x_{32}a_2. \]
Moreover, if no element in $A$ is dependent on the other two, then each $x_{ij} \ge 1$ and each $L_i=x_{ji}+x_{ki}$.  We have 
\[ \Ap(S,a_1) = \big\{ a_2x+a_3y: 0 \le x \le x_{12}-1, 0 \le y \le L_3-1 \;\;\text{or}\;\; 0 \le x \le L_2-1, 0 \le y \le x_{13}-1 \big\}. \]
\end{proposition}


\subsection{The case of compound sequences} \label{CS}

Let $a_1,\ldots,a_k$ and $b_1,\ldots,b_k$ be two sequences of positive integers such that $a_i<b_i$ for each $i$ and $\gcd(a_i,b_j)=1$ for each pair $i,j$, $i \ge j$. The {\it compound sequence\/} formed from these two sequences is 
\[ c_0 = a_1 a_2 a_3 \cdots a_k, c_1 = b_1 a_2 a_3 \cdots a_k, c_2 = b_1 b_2 a_3 \cdots a_k, \ldots, c_k = b_1 b_2 b_3 \cdots b_k. \]  



Note that $\gcd(c_0,c_1,c_2,\ldots,c_k)=1$. Two important special cases are 
\begin{itemize}
\item
The compound sequence for $a_1=\cdots=a_k=a$ and $b_1=\cdots=b_k=b$, $\gcd(a,b)=1$ is the geometric sequence 
\[ a^k, a^{k-1}b, a^{k-2}b^2, \ldots, b^k. \]
\item
For pairwise coprime positive integers $a_1,\ldots,a_k$, the compound sequence for $a_2,a_3,\ldots,a_k$ and $a_1,a_2,\ldots,a_{k-1}$ is the supersymmetric sequence 
\[ \frac{P}{a_1}, \frac{P}{a_2}, \ldots, \frac{P}{a_k}, \]
where $P=a_1 a_2 \cdots a_k$. 
\end{itemize}

Numerical semigroups generated by compound sequences were studied by Kiers et al.\ \cite{KNP16}. In their study, Kiers et al.\ determined an Ap\'{e}ry set, the Frobenius number, Betti elements, and catenary degree, and also computed bounds on the delta set. We use their result on the Ap\'{e}ry set of numerical semigroups to determine the set of fundamental gaps of these semigroups for the case where the embedding dimension equals $3$. 
 
\begin{lemma} {\bf \cite[Theorem 15]{KNP16}} \label{minclass_CS}
Let $a_1,\ldots,a_k$ and $b_1,\ldots,b_k$ be two sequences of positive integers such that $a_i<b_i$ for each $i$ and $\gcd(a_i,b_j)=1$ for each pair $i,j$, $i \ge j$. Then an Ap\'{e}ry set for the numerical semigroup $S$ generated by the compound sequence $c_0,c_1,\ldots,c_k$ of these two sequences is given by 
\[ \Ap(S,c_0) = \left\{ \sum_{i=1}^k c_ix_i: 0 \le x_i \le a_i-1, i=1,\ldots,k \right\}. \]
\end{lemma}

The compound sequence of $a_1,a_2$ and $b_1,b_2$ is a $3$-term sequence $c_0,c_1,c_2$. We study the set of fundamental gaps of numerical semigroups generated by $c_0,c_1,c_2$. 

\begin{theorem}
Let $a_1,a_2$ and $b_1,b_2$ be two sequences of positive integers such that $a_i<b_i$ for each $i$ and $\gcd(a_i,b_j)=1$ for each pair $i,j$, $i \ge j$. Let $c_0=a_1a_2$, $c_1=b_1a_2$, and $c_2=b_1b_2$, and let $S=\la c_0,c_1,c_2 \ra$. Let $b_2=qa_1+r$, $0 \le r<a_1$, and $\d=\left\lf \frac{2r}{a_1}\right\rf$. Let $n=c_1x+c_2y-c_0z \in \text{G}(S)$. 
\begin{itemize}
\item[{\rm (I)}]
\[ 2n \in S \LIff 
\begin{cases}
\begin{aligned}
\ts\frac{a_1}{2} & \le x < a_1,                                & 0 & \le y < \ts\frac{a_2}{2},       & 1 & \le z \le \ts\frac{b_1}{2},                 & \text{or} \\
0 & \le x < \ts\frac{a_1-r}{2},                                & \ts\frac{a_2}{2} & \le y < a_2,   & 1 & \le z \le \ts\frac{qb_1}{2},               & \text{or} \\
\ts\frac{a_1-r}{2} & \le x < \ts\frac{2a_1-r}{2}, & \ts\frac{a_2}{2} & \le y < a_2,   & 1 & \le z \le \ts\frac{(q+1)b_1}{2},         & \text{or} \\
\ts\frac{2a_1-r}{2} & \le x < a_1,                          & \ts\frac{a_2}{2} & \le y < a_2,   & 1 & \le z \le \ts\frac{(q+2)b_1}{2}.
\end{aligned}
\end{cases}
\]
\item[{\rm (II)}]
\[ 3n \in S \LIff 
\begin{cases} 
\begin{aligned} 
\ts\frac{a_1}{3} & \le x < \ts\frac{2a_1}{3},                              & 0 & \le y < \ts\frac{a_2}{3},                              & 1 & \le z \le \ts\frac{b_1}{3},   
& \text{or} \\
\ts\frac{2a_1}{3} & \le x < a_1,                                                    & 0 & \le y < \ts\frac{a_2}{3},                              & 1 & \le z \le \ts\frac{2b_1}{3}, 
& \text{or} \\
0 & \le x < \ts\frac{a_1-r}{3},                                                       & \ts\frac{a_2}{3} & \le y < \ts\frac{2a_2}{3},  & 1 & \le z \le \ts\frac{qb_1}{3}, 
& \text{or} \\
\ts\frac{a_1-r}{3} & \le x < \ts\frac{2a_1-r}{3},                         & \ts\frac{a_2}{3} & \le y < \ts\frac{2a_2}{3}, & 1 & \le z \le \ts\frac{(q+1)b_1}{3}, 
& \text{or} \\
\ts\frac{2a_1-r}{3} & \le x < \ts\frac{3a_1-r}{3},                       & \ts\frac{a_2}{3} & \le y < \ts\frac{2a_2}{3},  & 1 & \le z \le \ts\frac{(q+2)b_1}{3}, 
& \text{or} \\
\ts\frac{3a_1-r}{3} & \le x < a_1,                                                  & \ts\frac{a_2}{3} & \le y < \ts\frac{2a_2}{3},  & 1 & \le z \le \ts\frac{(q+3)b_1}{3}, 
& \text{or} \\
0 & \le x < \ts\frac{(1+\d)a_1-2r}{3},                                           & \ts\frac{2a_2}{3} & \le y <a_2,                         & 1 & \le z \le \ts\frac{(2q+\d)b_1}{3},  
& \text{or} \\
\ts\frac{(1+\d)a_1-2r}{3} & \le x < \ts\frac{(2+\d)a_1-2r}{3}, & \ts\frac{2a_2}{3} & \le y <a_2,                         & 1 & \le z \le \ts\frac{(2q+\d+1)b_1}{3}, 
& \text{or} \\
\ts\frac{(2+\d)a_1-2r}{3} & \le x < \ts\frac{(3+\d)a_1-2r}{3}, & \ts\frac{2a_2}{3} & \le y <a_2,                         & 1 & \le z \le \ts\frac{(2q+\d+2)b_1}{3}, 
& \text{or} \\
\ts\frac{(3+\d)a_1-2r}{3} & \le x <a_1,                                        & \ts\frac{2a_2}{3} & \le y <a_2,                         & 1 & \le z \le \ts\frac{(2q+\d+3)b_1}{3}.
\end{aligned}
\end{cases}
\]
\end{itemize}
\end{theorem}

\begin{proof}
By Lemma \ref{minclass_CS}, $n \in \text{G}(S)$ if and only if $n=c_1x+c_2y-c_0z$, where $0 \le x < a_1$, $0 \le y <a_2$, and $z \in \N$.  

\begin{itemize}
\item[{\rm (I)}]
We derive conditions on $n$ for which $2n \in S$ by considering five cases. The argument in each case is similar, so we present the first two cases in detail and give only the result in the remaining three cases. 

\begin{itemize}
\item[{\rm (i)}] 
For $0 \le x <\frac{a_1}{2}$ and $0 \le y <\frac{a_2}{2}$, $2n=2c_1x+2c_2y-2c_0z \notin S$, since $2c_1x+2c_2y \in \Ap(S,c_0)$. 

\item[{\rm (ii)}]
For $\frac{a_1}{2} \le x <a_1$ and $0 \le y <\frac{a_2}{2}$, $2n=c_1(2x-a_1)+2c_2y-c_0(2z-b_1)$ and $c_1(2x-a_1)+2c_2y \in \Ap(S,c_0)$. Therefore $2n \in S \Iff z \le \frac{b_1}{2}$. 

\item[{\rm (iii)}]
For $0 \le x <\frac{a_1-r}{2}$ and $\frac{a_2}{2} \le y <a_2$, we have $2n\in S \Iff z \le \frac{qb_1}{2}$.

\item[{\rm (iv)}] 
For $\frac{a_1-r}{2} \le x < \frac{2a_1-r}{2}$ and $\frac{a_2}{2} \le y <a_2$, we have $2n \in S \Iff z\le \frac{(q+1)b_1}{2}$. 

\item[{\rm (v)}]
For $\frac{2a_1-r}{2} \le x <a_1$ and $\frac{a_2}{2} \le y <a_2$, we have $2n \in S \Iff z \le \frac{(q+2)b_1}{2}$.
\end{itemize}

\item[{\rm (II)}]
We derive conditions on $n$ for which $3n \in S$ by considering eleven cases. Again, the argument in each case is similar, so we present the first two cases in detail and give only the result in the remaining nine cases. 

\begin{itemize}

\item[{\rm (i)}] 
For $0 \le x <\frac{a_1}{3}$ and $0 \le y <\frac{a_2}{3}$, $3n=3xc_1+3yc_2-3zc_0 \notin S$, because $3c_1x+3c_2y \in \Ap(S,c_0)$.

\item[{\rm (ii)}] 
For $\frac{a_1}{3} \le x <\frac{2a_1}{3}$ and $0 \le y <\frac{a_2}{3}$, $3n=c_1(3x-a_1)+3c_2y-c_0(3z-b_1)$ and $c_1(3x-a_1)+3c_2y \in \Ap(S,c_0)$. Therefore $3n \in S \Iff z \le \frac{b_1}{3}$. 

\item[{\rm (iii)}]
For $\frac{2a_1}{3} \le x <a_1$ and $0 \le y <\frac{a_2}{3}$, we have $3n \in S \Iff  z \le \frac{2b_1}{3}$. 

\item[{\rm (iv)}]
For $0 \le x <\frac{a_1-r}{3}$ and $\frac{a_2}{3} \le y <\frac{2a_2}{3}$, we have $3n \in S \Iff z \le \frac{qb_1}{3}$.

\item[{\rm (v)}]
For $\frac{a_1-r}{3} \le x <\frac{2a_1-r}{3}$ and $\frac{a_2}{3} \le y <\frac{2a_2}{3}$, we have $3n \in S \Iff  z \le \frac{(q+1)b_1}{3}$.

\item[{\rm (vi)}]
For $\frac{2a_1-r}{3} \le x <\frac{3a_1-r}{3}$ and $\frac{a_2}{3} \le y <\frac{2a_2}{3}$, we have $3n \in S \Iff  z \le \frac{(q+2)b_1}{3}$.

\item[{\rm (vii)}]
For $\frac{3a_1-r}{3} \le x <a_1$ and $\frac{a_2}{3} \le y <\frac{2a_2}{3}$, we have $3n \in S \Iff z \le \frac{(q+3)b_1}{3}$.

\item[{\rm (viii)}] 
For $0 \le x <\frac{(1+\d)a_1-2r}{3}$ and $\frac{2a_2}{3} \le y <a_2$, we have $3n \in S \Iff z \le \frac{(2q+\d)b_1}{3}$. 

\item[{\rm (ix)}] 
For $\frac{(1+\d)a_1-2r}{3} \le x <\frac{(2+\d)a_1-2r}{3}$ and $\frac{2a_2}{3} \le y <a_2$, we have $3n \in S \Iff z \le \frac{(2q+\d+1)b_1}{3}$.

\item[{\rm (x)}]
For $\frac{(2+\d)a_1-2r}{3} \le x <\frac{(3+\d)a_1-2r}{3}$ and $\frac{2a_2}{3} \le y <a_2$, we have $3n \in S \Iff z \le \frac{(2q+\d+2)b_1}{3}$. 

\item[{\rm (xi)}]
For $\frac{(3+\d)a_1-2r}{3} \le x <a_1$ and $\frac{2a_2}{3} \le y <a_2$, we have $3n \in S \Iff  z \le \frac{(2q+\d+3)b_1}{3}$. 

\end{itemize}
\end{itemize}
\end{proof}

\subsection{The case $A=\{a,b,c\}$, where $a \mid (b+c)$} \label{b+c}

\begin{lemma} {\bf \cite[Theorem 1]{Tri13}} \label{minclass_b+c}
Let $A=\{a,b,c\}$, where $\gcd(a,b,c)=1$ and $a \mid (b+c)$. Then the Ap\'{e}ry set for the semigroup $S=\la a,b,c \ra$ is given by 
\begin{align*} 
\Ap(S,a) & = \big \{ \min\{bx, c(a-x)\}: 0 \le x \le a-1 \big\} \\ 
                                        & =
\begin{cases} 
bx, & \text{if $0 \le x \le \frac{ac}{b+c}$}; \\ 
c(a-x), & \text{if $\frac{ac}{b+c} < x \le a-1$.}              
\end{cases}
\end{align*}
\end{lemma}

\begin{theorem}
Let $S=\la a,b,c \ra$, where $\gcd(a,b,c)=1$. If $a \mid (b+c)$ and $q=\left\lf \frac{ac}{b+c}\right\rf$, then $n \in \FG(S)$ if and only if 
\begin{itemize}
\item[{\rm (I)}]
$q<\frac{2a}{3}$ and
\begin{align*} 
n & =bx-az, & \ts\frac{q}{2} & < x < \ts\frac{a}{3}, & 1 & \le z \le \ts\frac{b+c}{a}x - \ts\frac{c}{2}, \\
\text{or} \;\; n & =bx-az, & \ts\frac{a}{3} & \le x < \ts\frac{a}{2}, & 1 & \le z \le \min \left\{ \ts\frac{b+c}{a}x - \ts\frac{c}{2}, \ts\frac{b}{3} \right\},  \\
\text{or} \;\; n & =bx-az, & \ts\frac{a}{2} & \le x < \ts\frac{a+q}{3}, & 1 & \le z \le \ts\frac{b}{3},  \\
\text{or} \;\; n & =bx-az, & \ts\frac{a+q}{3} & \le x \le q, & 1 & \le z \le \min \left\{ \ts\frac{b+c}{a}x - \ts\frac{2c}{3}, \ts\frac{b}{2} \right\},  \\ 
\text{or} \;\; n & =c(a-x)-az, & q & < x < \ts\frac{2a}{3}, & 1 & \le z \le \min \left\{ \ts\frac{b+2c}{2} - \ts\frac{b+c}{a}x, \ts\frac{c}{3} \right\},  \\ 
\text{or} \;\; n & =c(a-x)-az, & \ts\frac{2a}{3} & \le x \le \ts\frac{a+q}{2}, & 1 & \le z \le \ts\frac{b+2c}{2} - \ts\frac{b+c}{a}x. 
\end{align*} 
\item[{\rm (II)}]
$q \ge \frac{2a}{3}$ and
\begin{align*} 
n & =bx-az, & \ts\frac{q}{2} & < x < \ts\frac{a}{2}, & 1 & \le z \le \min \left\{ \ts\frac{b+c}{a}x-\ts\frac{c}{2}, \ts\frac{b}{3} \right\}, \\
\text{or} \;\; n & =bx-az, & \ts\frac{a}{2} & \le x < \ts\frac{a+q}{3}, & 1 & \le z \le \ts\frac{b}{3},  \\
\text{or} \;\; n & =bx-az, & \ts\frac{a+q}{3} & < x < \ts\frac{2a}{3}, & 1 & \le z \le \min \left\{ \ts\frac{b+c}{a}x-\ts\frac{2c}{3}, \ts\frac{b}{2} \right\}, \\
\text{or} \;\; n & =bx-az, & \ts\frac{2a}{3} & \le x \le q, & 1 & \le z \le \ts\frac{b}{2}, \\ 
\text{or} \;\; n & =c(a-x)-az, & q & < x \le \ts\frac{a+q}{2}, & 1 & \le z \le \ts\frac{b+2c}{2}-\ts\frac{b+c}{a}x. 
\end{align*} 
\end{itemize}
\end{theorem}

\begin{proof}
We assume, without loss of generality, that $b<c$. Let $q=\left\lf \frac{ac}{b+c} \right\rf$. Since $\left\lf \frac{ab}{b+c}\right\rf+\left\lf \frac{ac}{b+c}\right\rf=a-1$ or $a$ according as $(b+c) \nmid ac$ or $(b+c) \mid ac$, we have $q \ge \frac{a-1}{2}$ if $(b+c) \nmid ac$ and $q>\frac{a}{2}$ if $(b+c) \mid ac$. By Lemma \ref{minclass_b+c}, $n \notin S$ if and only if $n=bx-az>0$, $1 \le x \le q$, $z \ge 1$ or $n=c(a-x)-az>0$, $q+1 \le x \le a-1$, $z \ge 1$. 

\bigskip

\noindent {\sc Case} I. \big($q < \frac{2a}{3}$\big)
\begin{itemize}
\item[(i)]
For $1 \le x \le \frac{q}{2}$, ${\m}_{bx}=bx$ and ${\m}_{2bx}=2bx$. Hence $n=bx-az$, so that $2n<{\m}_{2bx}$. Therefore $n \notin \FG(S)$ in this case. 
\item[(ii)]
For $\frac{q}{2} < x < \frac{a}{3}$, ${\m}_{bx}=bx$, ${\m}_{2bx}=c(a-2x)$, and ${\m}_{3bx}=c(a-3x)$. Hence $n=bx-az$, and $2n \ge {\m}_{2bx}$ and $3n \ge {\m}_{3bx}$ translate to $z \le \frac{b+c}{a}x-\frac{c}{2}$ and $z \le \frac{b+c}{a}x-\frac{c}{3}$, respectively. Thus, 
$n \in \FG(S) \Iff z \le \ts\frac{b+c}{a}x - \frac{c}{2}$.    
\item[(iii)] 
For $\frac{a}{3} \le x < \frac{a}{2}$, ${\m}_{bx}=bx$, ${\m}_{2bx}=c(a-2x)$, and ${\m}_{3bx}=b(3x-a)$. Thus, $n \in \FG(S) \Iff z \le \min \left\{ \frac{b+c}{a}x - \frac{c}{2}, \frac{b}{3} \right\}$. 
\item[(iv)]
For $\frac{a}{2} \le x < \frac{a+q}{3}$, ${\m}_{bx}=bx$, ${\m}_{2bx}=b(2x-a)$, and ${\m}_{3bx}=b(3x-a)$. Thus, $n \in \FG(S) \Iff z \le \frac{b}{3}$.  
\item[(v)]
For $\frac{a+q}{3} \le x \le q$, ${\m}_{bx}=bx$, ${\m}_{2bx}=b(2x-a)$, and ${\m}_{3bx}=c\big(a-(3x-a)\big)$. Thus, $n \in \FG(S) \Iff z \le \min \left\{ \frac{b+c}{a}x - \frac{2c}{3}, \frac{b}{2} \right\}$. 
\item[(vi)]
For $q < x < \frac{2a}{3}$, ${\m}_{bx}=c(a-x)$, ${\m}_{2bx}=b(2x-a)$, and ${\m}_{3bx}=c\big(a-(3x-a)\big)$. Thus, $n \in \FG(S) \Iff z \le \min \left\{ \frac{b+2c}{2} - \frac{b+c}{a}x, \frac{c}{3} \right\}$. 
\item[(vii)]
For $\frac{2a}{3} \le x \le \frac{a+q}{2}$, ${\m}_{bx}=c(a-x)$, ${\m}_{2bx}=b(2x-a)$, and ${\m}_{3bx}=b(3x-2a)$. Thus, $n \in \FG(S) \Iff z \le \frac{b+2c}{2} - \frac{b+c}{a}x$.  
\item[(viii)]
For $\frac{a+q}{2} < x \le a-1$, ${\m}_{bx}=c(a-x)$ and ${\m}_{2bx}=c\big(a-(2x-a)\big)$. Hence $n=c(a-x)-az$, so that $2n<{\m}_{2bx}$. 
Therefore $n \notin \FG(S)$ in this case. 
\end{itemize}

\noindent {\sc Case} II. \big($q \ge \frac{2a}{3}$\big) There are seven subcases to consider, of which subcases (i), (iii), and (vii) are the subcases (i), (iv), and (viii) from Case I. We only consider the four remaining subcases, listed as (ii), (iv), (v), and (vi). 
\begin{itemize}
\item[(ii)]
For $\frac{q}{2} < x < \frac{a}{2}$, ${\m}_{bx}=bx$, ${\m}_{2bx}=c(a-2x)$ (since $q<2x<a$), and ${\m}_{3bx}=b(3x-a)$ (since $a<3x<a+q$). Hence $n=bx-az$, and $2n \ge {\m}_{2bx}$ and $3n \ge {\m}_{3bx}$ translate to $z \le \frac{b+c}{a}x-\frac{c}{2}$ and $z \le \frac{b}{3}$, respectively. Thus, $n \in \text{FG}(S) \Iff z \le \min \left\{ \frac{b+c}{a}x-\frac{c}{2}, \frac{b}{3} \right\}$. 
\item[(iv)]
For $\frac{a+q}{3} < x < \frac{2a}{3}$, ${\m}_{bx}=bx$ (since $\frac{2a}{3} \le q$), ${\m}_{2bx}=b(2x-a)$ (since $a<2x<a+q$), and ${\m}_{3bx}=c\big(a-(3x-a)\big)$ (since $a+q<3x<2a$). Hence $n=bx-az$, and $2n \ge {\m}_{2bx}$ and $3n \ge {\m}_{3bx}$ translate to $z \le \frac{b}{2}$ and $z \le \frac{b+c}{a}x-\frac{2c}{3}$, respectively. Thus, $n \in \text{FG}(S) \Iff z \le \min \left\{ \frac{b+c}{a}x-\frac{2c}{3}, \frac{b}{2} \right\}$. 
\item[(v)]
For $\frac{2a}{3} \le x \le q$, ${\m}_{bx}=bx$, ${\m}_{2bx}=b(2x-a)$, and ${\m}_{3bx}=b(3x-2a)$. Thus, $n \in \text{FG}(S) \Iff z \le \frac{b}{2}$.
\item[(vi)]
For $q < x \le \frac{a+q}{2}$, ${\m}_{bx}=c(a-x)$, ${\m}_{2bx}=b(2x-a)$, and ${\m}_{3bx}=b(3x-2a)$. Thus, $n \in \text{FG}(S) \Iff z \le \frac{b+2c}{2}-\frac{b+c}{a}x$. 
\end{itemize}
\end{proof}

\subsection{The case $A=\{a,b,c\}$, where $a \mid \lcm(b,c)$} \label{lcm}

\begin{lemma} {\bf \cite[Theorem 8]{Tri17}} \label{minclass_lcm}
Let $A=\{a,b,c\}$, where $\gcd(a,b,c)=1$ and $a \mid \lcm(b,c)$. Then the Ap\'{e}ry set for the semigroup $S=\la a,b,c \ra$ is given by 
\[ \Ap(S,a) = \big \{ bx+cy: 0 \le x \le s-1, 0 \le y \le r-1 \big\}, \]
where $r=\gcd(a,b)$ and $s=\gcd(a,c)$.   
\end{lemma}

\begin{theorem}
Let $S=\la a,b,c \ra$, where $\gcd(a,b,c)=1$ and $a \mid \lcm(b,c)$. Let $r=\gcd(a,b)$ and $s=\gcd(a,c)$. Then $bx+cy-az \in \FG(S)$ if and only if $z \ge 1$ and 
\begin{align*} 
& 0 \le x < \ts\frac{s}{3}, & \ts\frac{r}{2} & \le y < \ts\frac{2r}{3}, & z & \le \ts\frac{c}{3s}, \\  
\text{or}\:\: & \ts\frac{s}{3} \le x < \frac{s}{2}, & \ts\frac{r}{2} & \le y < \ts\frac{2r}{3}, & z & \le \min\{ \ts\frac{c}{2s}, \ts\frac{b}{3r}+\ts\frac{c}{3s} \},   \\  
\text{or}\:\: & 0 \le x < \ts\frac{s}{2}, & \ts\frac{2r}{3} &\le y < r, & z & \le \ts\frac{c}{2s}, \\   
\text{or}\:\: & \ts\frac{s}{2} \le x < \ts\frac{2s}{3}, & 0 &\le y < \ts\frac{r}{3}, & z & \le \ts\frac{b}{3r}, \\ 
\text{or}\:\: & \ts\frac{s}{2} \le x < \ts\frac{2s}{3}, & \ts\frac{r}{3} &\le y < \ts\frac{r}{2}, & z & \le \min\{ \ts\frac{b}{2r}, \ts\frac{b}{3r}+\ts\frac{c}{3s} \} \\
\text{or}\:\: & \ts\frac{2s}{3} \le x < s, & 0 &\le y < \ts\frac{r}{2}, & z & \le \ts\frac{b}{2r}, \\
\text{or}\:\: & \ts\frac{s}{2} \le x < \ts\frac{2s}{3}, & \ts\frac{r}{2} &\le y < \ts\frac{2r}{3}, & z & \le \ts\frac{b}{3r}+\ts\frac{c}{3s}, \\
\text{or}\:\:& \ts\frac{s}{2} \le x < \ts\frac{2s}{3}, & \ts\frac{2r}{3} &\le y < r, & z & \le \min\{ \ts\frac{b}{2r}+\ts\frac{c}{2s}, \ts\frac{b}{3r}+\ts\frac{2c}{3s} \}, \\
\text{or}\:\: & \ts\frac{2s}{3} \le x < s, & \ts\frac{r}{2} & \le y < \ts\frac{2r}{3}, & z & \le \min\{ \ts\frac{b}{2r}+\ts\frac{c}{2s}, \ts\frac{2b}{3r}+\ts\frac{c}{3s} \}, \\ 
\text{or}\:\: & \ts\frac{2s}{3} \le x < s, & \ts\frac{2r}{3} & \le y < r, & z &\le \ts\frac{b}{2r}+\ts\frac{c}{2s}. 
\end{align*} 
\end{theorem}

\begin{proof}
We make repeated use of Lemma \ref{minclass_lcm}. Note that $a=rs$ and that $n \notin S$ if and only if $n=bx+cy-az>0$, where $0 \le x \le s-1$, $0 \le y \le r-1$ and $z \ge 1$.  

To determine conditions on $x,y,z$ under which $2n \in S$, we consider nine cases with $(x,y) \in \big[\frac{\l s}{2},\frac{(\l+1)s}{2}\big) \times \big[\frac{\mu s}{2},\frac{(\mu +1)s}{2}\big)$, with $\l, \mu \in \{0,1\}$. Fix the ordered pair $(\l,\mu)$. Then 
\begin{equation} \label{lcm_2}
2n = b(2x-\l s)+c(2y-\mu r)+a\big(\ts\frac{\l b}{r}+\ts\frac{\mu c}{s}-2z\big) \in S \;\text{if and only if}\; 2z \le \ts\frac{\l b}{r}+\ts\frac{\mu c}{s}. \end{equation}

To determine conditions on $x,y,z$ under which $3n \in S$, we consider nine cases with $(x,y) \in \big[\frac{\l s}{3},\frac{(\l+1)s}{3}\big) \times \big[\frac{\mu s}{3},\frac{(\mu +1)s}{3}\big)$, with $\l, \mu \in \{0,1,2\}$. Fix the ordered pair $(\l,\mu)$. Then 
\begin{equation} \label{lcm_3} 
3n = b(3x-\l s)+c(3y-\mu r)+a\big(\ts\frac{\l b}{r}+\ts\frac{\mu c}{s}-3z\big) \in S \;\text{if and only if}\; 3z \le \ts\frac{\l b}{r}+\ts\frac{\mu c}{s}. \end{equation}

We must consider only those pairs $(\l,\mu) \in \{0,1\} \times \{0,1\}$ in eqn.~\eqref{lcm_2} for which $\frac{\l b}{r}+\frac{\mu c}{s} \ge 2$ and simultaneously only those pairs $(\l,\mu) \in \{0,1,2\} \times \{0,1,2\}$ in eqn.~\eqref{lcm_3} for which $\frac{\l b}{r}+\frac{\mu c}{s} \ge 3$.  Therefore $n=bx+cy-az \in \FG(S)$ according to Table \ref{lcm_table} below. 

\begin{table}[H] 
\centering
\begin{tabular}{|c||c|c|c|c|} \hline
\multirow{2}{*}{$\frac{2r}{3} \le y < r$} & $\min \big\{ \frac{c}{2s}, \frac{2c}{3s} \big\}$ & $\min \big\{ \frac{c}{2s}, \frac{b}{3r}+\frac{2c}{3s} \big\}$ & \multirow{2}{*}{$\min \big\{ \frac{b}{2r}+\frac{c}{2s}, \frac{b}{3r}+\frac{2c}{3s} \big\}$} & $\min \big\{ \frac{b}{2r}+\frac{c}{2s}, \frac{2b}{3r}+\frac{2c}{3s} \big\}$ \\  
& $=\frac{c}{2s}$ & $=\frac{c}{2s}$ & & $=\frac{b}{2r}+\frac{c}{2s}$ \\ \hline
\multirow{2}{*}{$\frac{r}{2} \le y < \frac{2r}{3}$} & $\min \big\{ \frac{c}{2s}, \frac{c}{3s} \big\}$ & \multirow{2}{*}{$\min \big\{ \frac{c}{2s}, \frac{b}{3r}+\frac{c}{3s} \big\}$} & $\min \big\{ \frac{b}{2r}+\frac{c}{2s}, \frac{b}{3r}+\frac{c}{3s} \big\}$ & \multirow{2}{*}{$\min \big\{ \frac{b}{2r}+\frac{c}{2s}, \frac{2b}{3r}+\frac{c}{3s} \big\}$} \\ 
& $=\frac{c}{3s}$ & & $=\frac{b}{3r}+\frac{c}{3s}$ & \\ \hline
\multirow{2}{*}{$\frac{r}{3} \le y < \frac{r}{2}$} & \multirow{2}{*}{\textbf{X}} & \multirow{2}{*}{\textbf{X}} & \multirow{2}{*}{$\min \big\{ \frac{b}{2r}, \frac{b}{3r}+\frac{c}{3s} \big\}$} & $\min \big\{ \frac{b}{2r}, \frac{2b}{3r}+\frac{c}{3s} \big\}$ \\ 
& & & & $=\frac{b}{2r}$ \\ \hline
\multirow{2}{*}{$0 \le y < \frac{r}{3}$} & \multirow{2}{*}{\textbf{X}} & \multirow{2}{*}{\textbf{X}} & $\min \big\{ \frac{b}{2r}, \frac{b}{3r} \big\}$ & $\min \big\{ \frac{b}{2r}, \frac{2b}{3r} \big\}$ \\
& & & $=\frac{b}{3r}$ & $=\frac{b}{2r}$ \\ \hline \hline
& $0 \le x < \frac{s}{3}$ & $\frac{s}{3} \le x < \frac{s}{2}$ & $\frac{s}{2} \le x < \frac{2s}{3}$ & $\frac{2s}{3} \le x < s$ \\ \hline
\end{tabular}
\caption{Upper bounds for $z$} \label{lcm_table}
\end{table}
\end{proof}

\section{The case of arithmetic progressions} \label{section=AP}

Numerical semigroups generated by arithmetic progressions have been extensively studied; see \cite{Bat58, Gra73, Rob56, Sel77, Tri94, Tri03}, for instance. By $\APP(a,d;k)$, we mean the $k$-term arithmetic progression $\{a,a+d,\ldots,a+(k-1)d\}$, with $\gcd(a,d)=1$ and $k \ge 2$. The Ap\'{e}ry set for the semigroup $S=\la\APP(a,d;k)\ra$ is given as Lemma \ref{minclass_AP}. We use this to determine the set of fundamental gaps in $S$ in Theorem \ref{AP}.        

\begin{lemma} {\bf \cite[Lemma 2]{Tri94}} \label{minclass_AP}
Let $a,d,k$ be positive integers, with $\gcd(a,d)=1$. Let $\APP(a,d;k)=\{a+id: 0 \le i \le k-1\}$. Then the Ap\'{e}ry set for the semigroup $S=\la  \APP(a,d;k)\ra$ is given by 
\[ \Ap(S,a) = \left\{ a\left(1+\left\lf\ts\frac{x-1}{k-1}\right\rf\right)+dx: 0 \le x \le a-1 \right\}. \]
\end{lemma}

\begin{theorem} \label{AP}
Let $a,d,k$ be positive integers, with $\gcd(a,d)=1$. Let $\APP(a,d;k)=\{a+id: 0 \le i \le k-1\}$. Then $ax+dy \in \FG\big(\APP(a,d;k)\big)$ if and only if 
\begin{align*} 
& & \ts\frac{1}{3} \left( \left\lf\ts\frac{3y-a}{k-1}\right\rf - (d-1) \right) & \le x \le \left\lf\ts\frac{y-1}{k-1}\right\rf, & \ts\frac{a}{2} & \le y < \ts\frac{2a}{3}, \\ 
\text{or} & & \ts\frac{1}{2}\left( \left\lf\ts\frac{2y-a}{k-1}\right\rf - (d-1) \right) & \le x \le \left\lf\ts\frac{y-1}{k-1}\right\rf, & \ts\frac{2a}{3} & \le y \le a-1.
\end{align*} 
\end{theorem}

\begin{proof}
Recall that $n$ is a fundamental gap of the numerical semigroup $S$ if and only if $n \notin S$ and $2n, 3n \in S$. Let $S=\la \APP(a,d;k) \ra$. We make repeated use of Lemma \ref{minclass_AP}. 

Fix $y \in \{1,\ldots,a-1\}$, and suppose $n=ax+dy>0$. Then $n \notin S$ if and only if $x \le \left\lf\frac{y-1}{k-1}\right\rf$. 

To determine conditions on $x$ under which $2n \in S$, we consider two cases: (i) $y \le \frac{a-1}{2}$, and (ii) $y>\frac{a-1}{2}$. In case (i), $2n=2ax+2dy \in S$ if and only if $2x \ge 1+\left\lf\frac{2y-1}{k-1}\right\rf$. Since $1+\left\lf\frac{2y-1}{k-1}\right\rf>2\left\lf\frac{y-1}{k-1}\right\rf$, the necessary and sufficient condition requires $x>\left\lf\frac{y-1}{k-1}\right\rf$, in contradiction to the requirement for $n \notin S$. Hence there is no fundamental gap in case (i). In case (ii), $y \ge \frac{a}{2}$, so $2n=a(2x+d)+d(2y-a) \in S$ if and only if $2x+d \ge 1+\left\lf\frac{2y-a}{k-1}\right\rf$.   

To determine conditions on $x$ under which $3n \in S$, we consider three cases: (i) $y \le \frac{a-1}{3}$, (ii) $\frac{a-1}{3} < y < \frac{2a}{3}$, and (iii) $\frac{2a}{3} \le y \le a-1$. In order that $n$ be a fundamental gap, we must have $y \ge \frac{a}{2}$. Hence there is no fundamental gap in case (i). In case (ii), $3n=a(3x+d)+d(3y-a) \in S$ if and only if $3x+d \ge 1+\lf\frac{3y-a}{k-1}\rf$. In case (iii), $3n=a(3x+2d)+d(3y-2a) \in S$ if and only if $3x+2d \ge 1+\left\lf\frac{3y-2a}{k-1}\right\rf$.  

Therefore $n=ax+dy>0$ is a fundamental gap for the set $S$ if and only if one of the following holds:
\begin{itemize}
\item
for $\frac{a}{2} \le y < \frac{2a}{3}$, 
\[ x \ge \max \left\{ \ts\frac{1}{2}\left( \left\lf\ts\frac{2y-a}{k-1}\right\rf - (d-1) \right), \frac{1}{3}\left( \left\lf\ts\frac{3y-a}{k-1}\right\rf - (d-1) \right) \right\}, \] 
\item
for $\frac{2a}{3} \le y \le a-1$, 
\[ x \ge \max \left\{ \ts\frac{1}{2}\left( \left\lf\ts\frac{2y-a}{k-1}\right\rf - (d-1) \right), \frac{1}{3}\left( \left\lf\ts\frac{3y-2a}{k-1}\right\rf - (2d-1) \right) \right\}. \] 
\end{itemize}

In the first case, we have 
\begin{align*}
2 \left( \left\lf\ts\frac{3y-a}{k-1}\right\rf - (d-1) \right) - 3 \left( \left\lf\ts\frac{2y-a}{k-1}\right\rf - (d-1) \right) & = 2 \left\lf\ts\frac{3y-a}{k-1}\right\rf  - 3 \left\lf\ts\frac{2y-a}{k-1}\right\rf + (d-1) \\
& \ge \left( \left\lf\ts\frac{2(3y-a)}{k-1}\right\rf - 1 \right) - \left\lf\ts\frac{3(2y-a)}{k-1}\right\rf + (d-1) \\
& \ge \left\lf\ts\frac{a}{k-1}\right\rf + (d-2) \:\text{since}\: \lf x \rf - \lf y \rf \ge \lf x-y \rf \\ 
& \ge 0. 
\end{align*}

In the second case, we have 
\begin{align*}
3 \left( \left\lf\ts\frac{2y-a}{k-1}\right\rf - (d-1) \right) - 2 \left( \left\lf\ts\frac{3y-2a}{k-1}\right\rf - (2d-1) \right) & = 3 \left\lf\ts\frac{2y-a}{k-1}\right\rf  - 2 \left\lf\ts\frac{3y-2a}{k-1}\right\rf + (d+1) \\
& \ge \left( \left\lf\ts\frac{3(2y-a)}{k-1}\right\rf - 2 \right) - \left\lf\ts\frac{2(3y-2a)}{k-1}\right\rf + (d+1) \\
& \ge \left\lf\ts\frac{3(2y-a)}{k-1}\right\rf - \left\lf\ts\frac{2(3y-2a)}{k-1}\right\rf  \\ 
& \ge 0. 
\end{align*}

Thus, $n=ax+dy>0$ is a fundamental gap if and only if
\begin{align*} 
& & \ts\frac{1}{3} \left( \left\lf\ts\frac{3y-a}{k-1}\right\rf - (d-1) \right) & \le x \le \left\lf\ts\frac{y-1}{k-1}\right\rf, & \ts\frac{a}{2} & \le y < \ts\frac{2a}{3}, \\ 
\text{or} & & \ts\frac{1}{2}\left( \left\lf\ts\frac{2y-a}{k-1}\right\rf - (d-1) \right) & \le x \le \left\lf\ts\frac{y-1}{k-1}\right\rf, & \ts\frac{2a}{3} & \le y \le a-1.
\end{align*} 
\end{proof}

\section{Acknowledgment}

The authors gratefully acknowledge the suggestions of the anonymous referee.  The authors also thank Aman Kumar for his assistance in editing this manuscript.  



\bigskip
\hrule
\bigskip

\noindent 2020 {\it Mathematics Subject Classification}: Primary 20M13; Secondary 20M14, 20M30. 

\noindent {\it Keywords:} numerical semigroup, fundamental gap, Ap\'{e}ry set.

\bigskip
\hrule
\bigskip

\vspace*{+.1in}
\noindent
Received  August 19 2020;
revised versions received  January 21 2021; January 24 2021;
January 25 2021.
Published in {\it Journal of Integer Sequences}, 
January 25 2021.

\bigskip
\hrule
\bigskip

\noindent
Return to
\htmladdnormallink{Journal of Integer Sequences home page}{https://cs.uwaterloo.ca/journals/JIS/}.
\vskip .1in


\end{document}

                                                                                

